Author’s response to reviews

Title: Morus alba L. suppresses the development of atopic dermatitis induced by the house dust mite in NC/Nga mice

Authors:

Hye-Sun Lim (qp1015@kiom.re.kr)

Version: 2 Date: 25 August 2013

Author’s response to reviews: see over
August 25, 2013

BMC Complementary & Alternative Medicine

Dear James Ellis Prozenko editor:

We have recently obtained significant data on atopic dermatitis of *Morus alba* L. through *in vitro* and *in vivo* experiments. The title of manuscript is ‘*Morus alba* L. suppresses the development of atopic dermatitis induced by the house dust mite in *NC/Nga mice*’. The manuscript consists of 16 pages of text (4,347 words), 5 figures.

The paper has been written according to the guidelines provided by the BMC Complementary & Alternative Medicine. I would appreciate very much if you could review it for publication in the BMC Complementary & Alternative Medicine. The manuscript is not being submitted in its entirety or in part to any other journal.

We thank you in advance for your concern.

Sincerely yours,

Hye-Sun Lim,
Korea Institute of Oriental Medicine
1672 Yuseong-daero, Yuseong-gu, Daejeon,
305-811, Korea (south)